Author's response to reviews

Title: An Open Access Database of Genome-wide Association Results

Authors:

Andrew D Johnson (johnsonad2@nhlbi.nih.gov)  
Christopher J O'Donnell (odonnellc@nhlbi.nih.gov)

Version: 5 Date: 29 December 2008

Author's response to reviews:

Dear editors and reviewers,

To address the minor revisions we have made alterations to Tables 1 and 2, the captions and at a few points in the manuscript text. All changes/additions are highlighted in yellow. The principal changes are:

1) Tables 1 and 2 are now sorted by LD-adjusted rankings rather than density rankings;

2) Table 1 includes bins down to the rank of 40, and Table 2 to the rank of 55, by density and/or LD-adjusted density. This resulted mainly in the addition of information for more bins to Table 2. In a few cases bins in previously listed in Table 2 are no longer listed;

3) The manuscript text (mainly captions and a few points in the discussion) have been changed to reflect these changes.

We appreciate the reviewers comments and the editors consideration of the revisions for rapid publication.

With best regards,

Andrew D. Johnson